Author’s response to reviews

Title: The effects of the Mulligan Sustained Natural Apophyseal Glide (SNAG) mobilisation in the lumbar flexion range of asymptomatic subjects as measured by the Zebris CMS20 3-D motion analysis system.

Authors:

Maria Moutzouri (moutzouri_marie@yahoo.com)
Evdokia Billis (evdokia.v.billis@postgrad.manchester.ac.uk)
Nikolaos Strimpakos (nikstrimp@teilam.gr)
Polixeni Kottioka (xennik190804@yahoo.gr)
Jacqueline A Oldham (jackie.oldham@man.ac.uk)

Version: 11 Date: 26 September 2008

Author’s response to reviews:

Dear Reviewers,

Thank you for accepting our manuscript for publication to BMC Musculoskeletal Disorders, I am attaching the manuscript with some minor typographical/grammar corrections.

Sincerely,

Evdokia Billis